Author's response to reviews

Title: Rationale and Study Design for a Randomised Controlled Trial to Reduce Sedentary Time in Adults at Risk of Type 2 Diabetes Mellitus: Project STAND (Sedentary Time AND Diabetes)

Authors:

Emma G Wilmot (emma.wilmot@uhl-tr.nhs.uk)
Melanie J Davies (Melanie.Davies@uhl-tr.nhs.uk)
Charlotte L Edwardson (C.L.Edwardson2@Lboro.ac.uk)
Trish Gorely (P.J.Gorely@Lboro.ac.uk)
Kamlesh Khunti (kk22@le.ac.uk)
Myra Nimmo (M.A.Nimmo@Lboro.ac.uk)
Thomas Yates (ty20@le.ac.uk)
Stuart J.H. Biddle (S.J.H.Biddle@Lboro.ac.uk)

Version: 5 Date: 4 December 2011

Author's response to reviews:

Typos corrected.